Author’s response to reviews

Title: Cementless short-stem total hip arthroplasty in the elderly patient. Is it a safe option? A prospective multicentre observational study

Authors:

Georgios Gkagkalis (georgios.gkagkalis@umontreal.ca)

Patrick Goetti (patrick.goetti@chuv.ch)

Sabine Mai (sabine.mai@vitos-okk.de)

Ingmar Meinecke (ingmar.meinecke@helios-gesundheit.de)

Naeder Helmy (nieder.helmy@spital.so.ch)

Dominique Bosson (dobosson@sunrise.ch)

Karl Philipp Kutzner (kkutzner@joho.de)

Version: 2 Date: 25 Mar 2019

Author’s response to reviews:

Dear Editors,

Dear Reviewers.

we herewith re-submit our revision of the manuscript for renewed review. We hope you will appreciate our improvements and kindly ask again for consideration for publication in your journal.

Thank you very much.

Best regards,

The authors
1. In your Ethics approval section, please clarify whether consent obtained from participants was written or verbal. If the consent was verbal please include reasoning for this and state if an IRB approved this method of obtaining consent.

The authors: Consent was obtained in written form of each patient before inclusion. We have clarified this in the Declaration section.

2. Please add a section "Consent for publication" in the Declarations. You may state "Not applicable" in this section as no personal information is provided in your manuscript.

The authors: We added the section "Consent for publication" in the Declarations and stated "Not applicable".

3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colors. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

The authors: We have uploaded only the final clean version. Figures have been uploaded as separate files.